



# Article Improved KNN Algorithm for Fine-Grained Classification of Encrypted Network Flow

# Chencheng Ma<sup>1,2</sup>, Xuehui Du<sup>1,2,\*</sup> and Lifeng Cao<sup>1,2</sup>

- <sup>1</sup> National Digital Switching System Engineering and Technological Research Center, Zhengzhou 450000, China; machencheng07@foxmail.com (C.M.); caolf302@sina.com (L.C.)
- <sup>2</sup> Zhengzhou Science and Technology Institute, Zhengzhou 450000, China
- \* Correspondence: dxh37139@sina.com

Received: 8 January 2020; Accepted: 11 February 2020; Published: 13 February 2020



**Abstract:** The fine-grained classification of encrypted traffic is important for network security analysis. Malicious attacks are usually encrypted and simulated as normal application or content traffic. Supervised machine learning methods are widely used for traffic classification and show good performances. However, they need a large amount of labeled data to train a model, while labeled data is hard to obtain. Aiming at solving this problem, this paper proposes a method to train a model based on the K-nearest neighbor (KNN) algorithm, which only needs a small amount of data. Due to the fact that the importance of different traffic features varies, and traditional KNN does not highlight the importance of different features, this study introduces the concept of feature weight and proposes the weighted feature KNN (WKNN) algorithm. Furthermore, to obtain the optimal feature set and the corresponding feature weight set, a feature selection and feature weight self-adaptive algorithm for WKNN is proposed. In addition, a three-layer classification framework for encrypted network flows is established. Based on the improved KNN and the framework, this study finally presents a method for fine-grained classification of encrypted network flows, which can identify the encryption status, application type and content type of encrypted network flows with high accuracies of 99.3%, 92.4%, and 97.0%, respectively.

**Keywords:** encrypted network flow classification; K-nearest neighbor algorithm; feature selection and weighted; fine-grained analysis; small training set

### 1. Introduction

Traffic-classification technology plays an important role in network security defense mechanisms. It is the basis for analyzing network traffic, detecting network anomalies, and balancing network load [1]. However, while traffic encryption is often used to protect information transmission, it also complicates the network traffic classification and analysis [2,3]. Nowadays, cyber attacks are usually implemented through encrypted traffic [4], and most of them are simulated as normal-application [5] or normal-content [6] network flows, which bypasses the network defense system and causes great damage. Thus, the fine-grained classification including the analysis of application and content types of encrypted traffic is an important research area [7].

Machine learning is a well-known method in the field of encrypted traffic classification [8]. But the machine-learning method needs a great amount of labeled data to train a model in terms of achieving fine-grained classification [9], and it is difficult to realize in an actual network for the reasons that labeled data are hard to obtain [10] and the model should be updated periodically for coping with concept drift [11,12].

Therefore, this study proposes a classification method using a model trained with a small amount of labeled data, which can achieve fine-grained and accurate classification of encrypted network flows.

The K-nearest neighbor (KNN) algorithm is widely used to train an accurate model based on a small training set [13,14]. Traditional KNN determines the label of new data according to the labels of the K-nearest data points. The point distance calculation is based on non-weighted feature values, which is not appropriate enough to be implemented in the classification of network flows. For one thing, the influences of different traffic features are different to the actual distinction of two data points [15], i.e., the essential features accurately describe the distinction while useless features mislead the classification results. For another, the importance of a feature is different for different classification purposes. Thus, features selection and feature weights setting are key parts of fine-grained classification of traffic based on KNN.

This study aims to promote the performance of traffic classification by improving traditional KNN. Considering the different effects of different features, this study introduces the concept of feature weight and proposes a weighted feature KNN (WKNN) algorithm. To obtain the optimal feature set and the corresponding feature weight set, a feature selection and feature weight self-adaptive algorithm for WKNN (WKNN-Selfada) is proposed, which can be used to train a classification model for encrypted traffic identification. WKNN-Selfada can adjust weights according to each misleading sample, so it can fully learn the characteristics of the traffic just with a few training samples, which can meet the requirements of a small training set.

Furthermore, a framework for fine-grained classification of encrypted network flows is built, which analyzes three attributes of encrypted network flows, namely encryption status, encrypted application type, and encrypted content type. Fine-grained and multi-attribute classification is the basis of network management and network security analysis. Based on the framework, classification of network flows can be more meticulous, which provides many network analysis work with basic supports. For better understanding, we take a simple example. The framework can be used to analyze whether a flow is abnormal not only according to the single attribute value of a flow, but also the association between the attributes of the flow. For example, YouTube flow is normal flow in most cases and file flow is a normal flow in most cases, but a YouTube and file flow is likely a malicious flow, because the YouTube application usually produces streaming flows but hardly file flows. Thus, if we detect a flow as a YouTube flow as well as a file flow, the flow will be probably an anomaly. The framework we proposed can distinguish this type of abnormal flows based on the analysis of the correlation between the attributes.

Finally, based on the improved KNN algorithm and the framework, a new method for fine-grained classification of encrypted network flows (FCE-KNN) is proposed. This method implements the WKNN-Selfada algorithm to train classification models and uses these models for real-time traffic classification based on WKNN.

Th main contributions of this study are as follows:

(1) Aiming at solving the problem that different influences of features are not expressed accurately, improved versions of the traditional KNN algorithm are developed, namely the weighted feature KNN (WKNN) algorithm and the feature selection and feature weight self-adaptive algorithm for WKNN (WKNN-Selfada).

(2) To meet the requirement of network security analysis, a three-layer framework for fine-grained classification of an encrypted network flow is innovatively proposed, which can reinforce network security by analyzing the correlation of the fine-grained attributes.

(3) In order to realize accurate and fine-grained classification, a fine-grained classification of encrypted network flows based on the framework and the improved KNN algorithms (FCE-KNN) are presented, which can identify the encryption status, application type and content type of encrypted network flows.

The remainder of this paper is organized as follows. Section 2 reviews related studies on traffic classification. Section 3 introduces the notion of feature weight and proposes the WKNN and WKNN-Selfada algorithms. Section 4 discusses a fine-grained classification framework and proposes a corresponding fine-grained classification method for encrypted network flows. Section 5 presents

the results of experiments on the public dataset Information Security Centre of Excellence (ISCX) VPN-nonVPN [16] (VPN: Virtual Private Network, one of the means of encrypting network traffic) to compare the performance of FCE-KNN with that of state-of-the-art algorithms. Finally, Section 6 summarizes the main findings of the paper.

#### 2. Related Work

This study categorizes traffic-classification techniques from the perspective of extracted traffic features, as shown in Figure 1. The traffic-classification technology consists of static feature analysis and dynamic feature analysis. Static feature analysis mainly includes port detection and packet load detection, whereas dynamic feature analysis mainly includes statistical feature analysis and behavior analysis. Statistical feature analysis includes machine-learning methods and general statistical feature methods.



Figure 1. Categorization of traffic classification methods.

The port-detection technology implements traffic classification by detecting the port number of packets, but it is not always useful owing to the abuse of the non-standardized port number. Deep packet inspection (DPI) [17,18] and deep flow inspection (DFI) [19] implement analysis on packets or flows through frequent item-mining and pattern-matching methods, but it is difficult to establish matching rules for encrypted traffic. The behavior detection method [20–24] achieves high identification accuracy of encrypted traffic only for some special applications or protocols. The methods based on statistical features currently show relatively good performance for encrypted traffic classification, and machine learning (ML) is the most popular and effective one among them.

From the perspective of basic ML methods, decision trees [25,26] showed good performance in most cases, but over-fitting can easily occur, leading to poor classification in actual networks. Cluster methods [27,28] could achieve traffic classification without labeled data, but it was difficult to find a really useful and suitable clustering metrics. Bayesian methods [29,30] provided probability-based predictions, but their performance was poor when there were correlations between features, as was usually the case. Support vector machines (SVM) [31] had general adaptability to encrypted traffic classification, but the algorithm had high complexity and was very time consuming. Although ensemble learning [32,33] showed good performance, the model based on a weak learner lacked interpretability. Neural networks [34–43] needed a large amount of data to train a model, which was difficultly to realize in a condition of a small training set. Transfer learning [44] and active learning [45] addressed the issues of model practicability and insufficient label data during training, respectively, but they have not been adequately explored in studies on fine-grained classification. Various machine-learning methods had been combined to build hybrid models [46–48] in progressive or parallel structures [49], but most of the models were too complex for real-time classification. For the general statistical feature methods, Dorfinger et al. [50] identified encrypted traffic according to the entropy of packet data; however, some studies [1,29] have recently found that the entropy method cannot distinguish between encrypted traffic and compressed traffic. Among them, KNN [51–53] is light and accurate, and could train a high-performance model with a small amount of labeled data. But to some extent, the non-weight distance calculation lacks adaptability to different tasks of encrypted traffic

classification. Carela-Español et al. [54] used KD-Tree (k-dimension tree) to improve the efficiency of KNN in traffic classification. Bar-Yanai et al. [55] combined KNN and K-means to improve the efficiency for real-time traffic classification. However, These two methods did not solve the problem of fine-grained classification.

From the perspective of methods including feature selection or feature weight training, McGaughey et al. [56] used the fast orthogonal search algorithm for traffic feature selection. Dong et al. [57] proposed a multi-objective adaptive feature selection algorithm for traffic classification based on information gain rate and evolutionary computing. Saber et al. [58] achieved traffic feature selection based on linear discriminant analysis. Manju et al. [59] ranked traffic features according to the feature weights which were the number of times that each feature appears in the tree, and selected the optimal feature subset based on the accuracies of the extreme gradient boosting model. Jamil et al. [60] created several candidate feature subsets by different feature selection algorithms, and chose the best subset according to the results of all the feature subsets' evaluations based on the five ML algorithms. The feature weighted or feature selection method based on KNN [61] has been studied comprehensively in many areas, such as transportation system [62], anomaly detection [63,64], image identification [65] and so on, except for the fine-grained classification of network traffic. Only Dong et al. [66] in 2017 presented a modified version of consistency-based method in combination with a layered KNN classifier to evaluate the goodness of a feature subset.

#### 3. Improvement of K-Nearest Neighbor (KNN) Algorithm

This section introduces the notion of feature weight, improves the distance calculation formula in KNN, and proposes the weighted feature KNN (WKNN). Furthermore, this section also includes our proposed feature selection and feature weight self-adaptive algorithm for WKNN (WKNN-Selfada), considering how to choose the appropriate feature set and the optimal feature weight set.

#### 3.1. Weighted-Feature KNN

KNN is a supervised machine learning algorithm that finds a similarity between two points by calculating the distance between them. KNN first calculates the distances between each training sample and the target point, and then selects the k-nearest samples to the target point. These k samples jointly determine the class of the target point. The distance calculation of features is a direct means for expressing the similarity of points, and KNN shows excellent performance in predicting the target network flow.

Minkowski distance is one of the widely used distance metrics in traditional KNN, shown in below. If traffic features have no scale and have the same data distribution, Minkowski distance can express the actual distance between two points. In this paper, the exponent of the differences between the feature values is set to 2, which is Euclidean distance, because square sum is beneficial to describe the multi-dimension point distance. When the feature dimension of points is *m*, the formula for calculating the point distance between training sample  $p = (x_1, \ldots, x_m)$  and test point  $q = (y_1, \ldots, y_m)$  is given by *distance* =  $\sqrt{\sum_{j=1}^m (x_j - y_j)^2}$ . The shorter the distance between the two points, the greater

is given by *alstance* =  $\sqrt{\sum_{j=1}^{j} (x_j - y_j)}$ . The shorter the distance between the two points, the greater is the similarity between them. After calculating the distances of all point pairs, the k-nearest training samples determine the class of the test point.

The distance calculation of traditional KNN is non-weighted, which means that it does not reflect the different effects of different features in traffic classification. This paper introduces feature weight and weighted feature-based point distance to improve the adaptation of the classification model to encrypted traffic-classification tasks. We use data normalization to eliminate the data scales for the implementation of Minkowski distance. Based on Euclidean distance, we make improvements on the traditional KNN, and several definitions are shown below. **Definition 1** (Feature weight). *is the coefficient of the feature when calculating the distance between two points. The weights of features are represented by a column vector,*  $w = (w_1; ...; w_m)$ *, where m is the dimension of the features.* 

Feature weights reflect the importance of features during distance calculation. The larger the feature weight, the greater is the influence of the feature on the traffic classification.

**Definition 2** (Feature distance). *is the square of the difference between two points' feature of the dimension.* For points  $p = (x_1, ..., x_m)$  and  $q = (y_1, ..., y_m)$ , the feature distances are calculated and represented by  $fd = [(p - q).]^2 = ((x_1 - y_1)^2, ..., (x_m - y_m)^2)$ , where  $[(p - q).]^2$  means squaring each elements of the vector (p - q). If there are several pairs of two points, the feature distances would be formed a matrix, represented by **FD**, and  $fd^{(i)}$  means the ith pair's feature distances.

**Definition 3** (Weighted feature-based point distance). *is based on the Minkowski distance with the additional introduction of the feature weight. For points p, q and the feature weights w, the weighted feature-based point distance is given by distance = \sqrt{fd \cdot w} = \sqrt{[(p - q).]^2 \cdot w}.* 

The weighted-feature KNN (WKNN) algorithm is proposed in this paper, shown in Algorithm 1. The inputs of the algorithm are the training sample set, a matrix **P**, target point *q*, parameter *k*, and feature weight *w*. The shape of the matrix **P** is (n, m), where *n* is the number of training points and *m* is the feature dimension, and  $p^{(i)}$  means the *i*th row of **P**. The outputs are the class prediction result of the target point *q*, the weighted feature-based point distances between the k-nearest neighbors and *q*, *kDistances*, the matrix of feature distances between each k-nearest neighbor and *q*, **KFD**, and the classes of the k-nearest neighbors, *kClass*. Some of the outputs are used for Algorithm 2, which will be described later.

Algor	ithm 1: WKNN
	Data: P, q, w, k
	<b>Result</b> : classPrediction( <i>q</i> ), <i>kDistances</i> , <b>KFD</b> , <i>kClass</i>
1	initialize $distances \leftarrow \{0\}, FD \leftarrow \{0\}$
2	for $i = 1$ to len(P) do
3	$fd^{(i)} \leftarrow [(p^{(i)} - q).]^2$
4	$distances[i] \leftarrow \sqrt{fd^{(i)} \cdot w}$
5	end
6	$knnIndex \leftarrow argminK(distances, k)$
7	$\mathbf{KFD} \leftarrow (fd^{(knnIndex[0])}; \dots; fd^{(knnIndex[k-1])})$
8	$kDistances \leftarrow \{distances[knnIndex[0]], \dots, distances[knnIndex[k-1]]\}$
9	$kClass \leftarrow \{class(p^{(knnIndex[0])}), \dots, class(p^{(knnIndex[k-1])})\}$
10	$score \leftarrow \{0\}$
11	for $i = 1$ to k do
12	<pre>score[kClass[i]] += 1 / kDistance[i]</pre>
13	end
14	$classPrediction(q) \leftarrow argmax(score)$

First, in lines 1–5, the algorithm calculates the feature distances between each training point  $p^{(i)}$  and q, and then calculates the corresponding weighted feature-based point distance, where *distances*[*i*] means the *i*th pair's weighted feature-based point distance and *distances* is an array.

Next, in lines 6–9, the algorithm calculates the information of the k-nearest neighbors. *knnIndex* is the corresponding indexes of the k-nearest neighbors. *kDistances* is the weighted feature-based point

distances of the k-nearest neighbors. **KFD** is the feature distances matrix of the k-nearest neighbors. *kClass* is the labels of the k-nearest neighbors. argminK() is a function to get the indexes of the *k* smallest elements from small to large, and class() is to get the label of the point.

Finally, in lines 10–14, it calculates the prediction scores. WKNN uses the reciprocal of each k distance value as a weight to be added to the score of the corresponding class and output the class with the highest score as the prediction result.

## 3.2. Feature Selection and Feature Weight Self-Adaptive Algorithm for Weighted Feature KNN (WKNN)

The selection of the feature set and the setting of the corresponding feature weights determine whether the calculated point distance can accurately represent the similarity between two points. To achieve more effective classification, a feature selection and feature weight self-adaptive algorithm for WKNN (WKNN-Selfada) is proposed, which can adapt feature weights by itself based on training data, instead of by manual setting. The algorithm learns the influence and updates the weights of features by analyzing only one sample at a time, so it can fully learn and adapt well to the law of each training sample and realize accurate classification just with a small training set.

The algorithm includes two parts, which runs two times and achieves a single part each time. The first time of the algorithm adapts the feature weight of each feature in the candidate feature set and selects the optimal feature set by comparing the weights with the feature selection threshold. In the second time of the algorithm, after feature selection in the first time, the weights of the selected features are retrained, because the new feature set is not the same with the original feature set and the weights of the original features cannot express the actual influences and mutual relation of the newly selected features. In the process of weight adjustment, the algorithm suggests that the feature with a larger feature distance would play a greater role in distinguishing the two points with different classes once misclassification. Therefore, when updating the weights, the weights of features with a large feature distance are supposed to increase while those with a small feature distance are supposed to decrease in order to increase the point distance between two points with different classes and reduce the possibility of misclassification. Thus, the accuracy of identifying the class of the test point can be improved further.

Unlike the traditional KNN algorithm, the improved algorithm presented in this paper involves a training process. Therefore, not only to compare with the target point for distance calculation, the training data needs to be divided for updating the weights. For clarity, the two training sets are called the decision samples set and the weights update set.

As shown in Algorithm 2, there are several input parameters: *trainData* is the data used for training, *k* is the k value of WKNN, *nRound* is the number of rounds for training, *raDiv* is the ratio for dividing the training samples into the decision samples set and the weights update set, and  $\delta$  is an adjustment parameter used to calculate the feature selection threshold. It finally outputs the indexes of the selected features and the corresponding feature weights. The algorithm runs two times automatically by judging whether the value of  $\delta$  is null. The first time is to select a feature set with no null  $\delta$  and the second is to train the feature weights for the new features with null  $\delta$ .

First, in lines 1–3, WKNN-Selfada initializes the selected feature index set, *selectedFeatureIndex*, as all feature indexes, i.e.,  $\{0, ..., m-1\}$ . Then it uses the function DataProcessing() to extract the feature values of the data according to the selected feature and initializes the feature weights, *w*, as values of 1/m, where *m* is the feature dimension.

Next, in lines 4–22, it loops several times at the value of *nRound*. At the beginning of each round, the method starts with a data division function, dataDivide(), which proportionally divides *trainData* into a decision samples set **P** and a weights-update set **Q** in the ratio *raDiv*. One training round involves multiple instances of training, which is equal to the size of the weights-update set **Q**. Line 7 implements the WKNN algorithm on decision sample set **P** and the weights-update sample *q* and yields the class prediction named *prediction*, the weighted feature-based point distances of the k-nearest decision samples, *kDistances*, the matrix of feature distances between each k-nearest decision

samples and *q*, **KFD**, and the classes of the k-nearest decision samples, *kClass*. Then, it judges whether the prediction class is true. If false, it then implements the feature weights-update, which judges in sequence whether the class of each nearest k decision sample is equal to the true class of target point *q*. If not, the algorithm starts to update the weights, as shown in lines 11–17. In the phase of weights update, the process in the class of the *i*th k-nearest decision sample is different from the target sample, which is described below.

Algorithm 2: WKNN-Selfada								
<b>Data</b> : trainData, k, nRound, raDiv, $\delta$								
<b>Result</b> : selectedFeatureIndex, <i>w</i>								
1 initialize $m_0 \leftarrow$ the length of trainData's feature dimension, <i>selectedFeatureIndex</i> $\leftarrow \{0,, m_0 - 1\}$								
2 initialize <i>trainData</i> ← dataProcessing( <i>trainData</i> , <i>selectedFeatureIndex</i> )								
3 initialize $\beta = m \leftarrow \text{len}(\text{selectedFeatureIndex}), \alpha_1 = \cdots \alpha_m \leftarrow 1, w \leftarrow (\alpha_1/\beta; \ldots; \alpha_m/\beta)$								
4 for $r = 1$ to <i>nRound</i> do								
5 <b>P</b> , <b>Q</b> $\leftarrow$ dataDivide( <i>trainData</i> , <i>raDiv</i> )								
6 <b>for</b> each $q$ in $\mathbf{Q}$ do								
7 prediction, kDistances, <b>KFD</b> , kClass $\leftarrow$ WKNN( <b>P</b> , <i>q</i> , <i>w</i> , <i>k</i> )								
8 <b>if</b> prediction is not class(q) <b>then</b>								
9 for $i = 1$ to k do								
10 if $kClass[i]$ is not class(q) then								
11 $rank \leftarrow getFeatureRank(kfa^{(r)})$								
$\frac{12}{\lambda \leftarrow \min(kDistances[i])}$								
$\frac{13}{\beta \leftarrow \beta + \lambda \cdot m \cdot (m+1)/2}$								
14 for $j = 1$ to $m$ do								
15 $\alpha_j \leftarrow \alpha_j + \lambda \cdot rank[j]$								
16 end								
17 $w \leftarrow \{\alpha_1/\beta, \dots, \alpha_m/\beta\}$								
18 end								
19   end								
20 end								
21   end								
22 end								
23 if $\delta \neq \text{null } \mathbf{do}$								
$\frac{24}{10}  \text{for } j = 1 \text{ to } m \text{ do}$								
25 if $w_j < 1/m + \delta \operatorname{std}(w)$								
26 delete <i>j</i> from <i>selectedFeatureIndex</i>								
27 end								
28 end								
29 $\delta = \text{null}$								
30   goto line 2								
31 end								

- Step 1: obtain the ranks of each feature based on the feature distances  $kfd^{(i)}$ . For example, if  $kfd^{(i)} = (0.9, 0.2, 0.4)$ , the ranks of the features are (3,1,2), where the *i*th element of *rank* means the rank of the *i*th feature according to the feature distance from small to large.
- Step 2: obtain the parameter λ, which is the update ratio of the weights. It is set to the ratio of the smallest weighted feature-based point distance in this loop, min(*kDistances*), to the weighted feature-based point distance between this decision sample and the target sample, *kDistances*[*i*].

- Step 3: calculate the denominator of the new weights,  $\beta$ . The denominator increment is given by  $\Delta\beta = \lambda \cdot \frac{m(m+1)}{2}$ , where  $\lambda$  is the update ratio and m is the dimension of the feature vector. Thus, the new denominator is  $\beta \leftarrow \beta + \Delta\beta$ .
- Step 4: calculate the molecular of the new weights,  $\alpha$ . The molecular increments are given by  $\Delta \alpha_i = \lambda \cdot rank[i]$ , where *i* is the feature index. Thus, the new molecular are  $\alpha_i \leftarrow \alpha_i + \Delta \alpha_i$ , where  $\alpha_i$  is the weight's molecular of the *i*th feature.
- Step 5: calculate the new feature weights,  $w = (w_1; ...; w_m) \leftarrow \left(\frac{\alpha_1}{\beta}; ...; \frac{\alpha_m}{\beta}\right)$ . The sum of all weights equal to 1 under any situation.

The misclassified decision sample at a shorter distance from the test point produces a larger update ratio  $\lambda$ , which means that it has a greater influence on the training process for feature weight update. After additional rounds of processing the feature weight self-adaptively, each feature weight converges to a certain value. In particular, if the influence of one feature is extremely low (or extremely high), the weight will converge to  $2/[m \cdot (m+1)]$  (or 2/(m+1)).

Then, in lines 23–31, the algorithm determines whether feature selecting is finished by judging whether  $\delta$  is null. As  $\delta$  does not have a null value in the first iteration, the processing of feature selection would be performed, i.e., lines 24–28. The algorithm selects the features by judging whether the weight of each feature is smaller than the feature selection threshold, ranging from the 1st dimension to the *m*th dimension in order. If so, the feature index corresponding to the weight is removed from *selectedFeatureIndex*. The threshold is obtained by calculating the difference between the average value of the weights, 1/m and  $\delta$  times the standard deviation of the weights. Here,  $\delta$  is an adjustable parameter and std() is the standard deviation function. The basic idea behind the threshold design is that the feature weights obey normal distribution, so values lower than the confidence intervals are rare and have low influences on the classification. In statistics, once the mean and standard deviation of the data are given, the  $\delta$  can be determined based on the confidence level such as 95%, and the confidence intervals can be calculated, so values lower than the intervals can be removed. However, WKNN-Selfada does not use the notion of confidence level, but sets the  $\delta$  and the threshold by experimental verification instead. At the end of the first iteration, the algorithm sets the  $\delta$  to null and then begins the second iteration from line 2. In the second iteration, it calculates the new weights of the new feature according to the *selectedFeatureIndex*, shown in lines 4–22. After finishing the weight update, the algorithm ends up in the  $\delta$  of null value, and outputs the feature indexes of the new feature set, *selectedFeatureIndex*, and the corresponding new feature weights, *w*.

#### 4. Fine-Grained Classification of Encrypted Network Flows

This section proposes a fine-grained classification framework for encrypted traffic classification, which includes network flow division, flow feature extraction, fined-grained label designation, model training, and real-time classification. Based on this framework and the improved KNN algorithms presented in Section 3, a new method is proposed to realize fine-grained classification of encrypted network flows.

#### 4.1. Description of Classification Framework

The main idea of the framework for fine-grained classification of encrypted network flows is to train three classifiers using the hierarchical features and classes of sample traffic data and use them to predict the fine-grained labels of real-time network flows. The framework is shown in Figure 2.



Figure 2. Framework of fine-grained real-time classification for encrypted network flows.

The framework consists of two parts: offline training and online classification. The offline part selects the hierarchical features and trains the corresponding feature weights of the three-layer model on the basis of the WKNN-Selfada algorithm after the processe of flow division and class extraction. The online part divides the flows from real-time traffic, extracts the flow features, and calculates the fine-grained classification result on the basis of the training model. The classification model calculates the point distances using the WKNN algorithm.

The key aspect of the framework is the fine-grained three-layer classifier model. The first layer is the traffic encryption status identification layer, which is the basis of the entire framework. It identifies an encrypted flow through the feature set of the encryption status. The identified encrypted flow then undergoes fine-grained analysis, i.e., it enters the second and third layers of the model. The second layer is the application identification layer for encrypted flows. The feature set of application type is used to identify the application to which the encrypted network flow belongs. The third layer is the content type identification layer. The feature set of content type is used to identify the content type (such as files, simple communication messages, audio and video communication, or multimedia streams) transmitted by the encrypted application flow, which undergoes further fine-grained analysis for the encrypted flow.

Combining with the analysis of the correlations of the fine-grained results and the secure rules, a more effective network defense would be implemented. For example, a malicious flow simulated as the YouTube flow, is detected as a YouTube and file flow by the framework. If we just analyze a single attribute value of the flow, e.g., YouTube or file, the malicious flow would be regarded as normal flow, which would cause damage to the network. However, if we analyze the correlations of the attributes and match the secure rule such as "If a YouTube flow is a file flow, the flow is suspicious.", the malicious flow would attract the network manager's attention and further be prevented from damage.

#### 4.2. Design of Candidate Feature Set

The hierarchical feature set of the framework is based on the feature selection part of the WKNN-Selfada algorithm, which selects features from the candidate feature set shown in Table 1. In the field of traffic classification, traffic features can be classified as packet features and flow features. Packet feature extraction is relatively simple and efficient, but its classification accuracy is low. Furthermore, the analysis of flow features is complicated. However, as network flows can be regarded as basic units of network behavior between pair-wise subjects, the analysis based on flow features is more

\_

comprehensive in the area of network traffic and it achieves higher classification accuracy. The definition of network flow is as follows. In this paper, flow refers to bilateral flow unless otherwise specified.

	Feature	Calculation Formula
1	Minimum of packet inter-arrival time	$iat_min = min(iat[])$
2	Maximum of packet inter-arrival time	$iat_max = max(iat[])$
3	Mean of packet inter-arrival time	$iat\_mean = mean(iat[])$
4	Standard deviation of packet inter-arrival time	$iat\_std = std(iat[])$
5	Minimum of IP packet bytes	$byte\_min = min(byte[])$
6	Minimum of IP packet bytes	$byte_max = max(byte[])$
7	Mean of IP packet bytes	<i>byte_mean</i> = mean( <i>byte</i> [])
8	Standard deviation of IP packet bytes	$byte\_std = std(byte[])$
9	Number of IP packet bytes per second	$byte\_psec = \frac{\sum_{i=1}^{n} byte[i]}{\sum_{i=1}^{n} iat[i]}$
10	Number of packets per second	$pac\_psec = \frac{n}{\sum_{i=1}^{n} iat[i]}$
11	Minimum of forward packet inter-arrival time	$fiat\_min = \min(fiat[]),$ fiat = {iatsd[i] d[i] == 0}[]
12	Maximum of forward packet inter-arrival time	$fiat_max = max(fiat[])$
13	Mean of forward packet inter-arrival time	$fiat\_mean = mean(fiat[])$
14	Standard deviation of forward packet inter-arrival time	$fiat\_std = std(fiat[])$
15	Minimum of IP packet bytes of forward packets	$fbyte\_min = min(fbyte[]),$
10	Wining of the pucket by tes of forward puckets	fbyte = {byte $[i] d[i] == 0$ }[]
16	Minimum of IP packet bytes of forward packets	$fbyte_max = max(fbyte[])$
17	Mean of IP packet bytes of forward packets	fbyte_mean = mean(fbyte[])
18	Standard deviation of IP packet bytes of forward packets	$fbyte\_std = std(fbyte[])$
19	Number of IP packet bytes of forward packets per second	$fbyte\_psec = \frac{\sum_{i=1}^{jp} fbyte[i]}{\sum_{i=1}^{n_{fp}} fiat[i]}$
20	Number of forward packets per second	$fpac\_psec = \frac{-\frac{n_{fp}}{n_{fp}}}{\sum_{i=1}^{n_{fp}} fiat[i]}$
21	Minimum of backward packet inter-arrival time	$biat_min = min(biat[]),$ $biat = {iatsd[i] d[i] == 1}[]$
22	Maximum of backward packet inter-arrival time	$biat_max = max(biat[])$
23	Mean of backward packet inter-arrival time	$biat_mean = mean(biat[])$
24	Standard deviation of backward packet inter-arrival time	$biat\_std = std(biat[])$
25	Minimum of IP packet bytes of backward packets	$bbyte_min = min(bbyte[]),$
26	Minimum of ID nocket bytes of backward nackets	$bbyte = \{bbyte[i]   a[i] == 1\}[]$
20	Maan of IP nacket bytes of backward packets	$bbyte_max = max(bbyte[])$
27	Standard deviation of IP packet bytes of backward packets	hute std - std(hhute[])
20	Sumare deviation of it packet by its of backware packets	$\sum_{i=1}^{n_{bp}} hhut_{e}[i]$
29	Number of IP packet bytes of backward packets per second	$bbyte\_psec = \frac{\sum_{i=1}^{n} bbyte[i]}{\sum_{i=1}^{n_{bp}} fiat[i]}$
30	Number of backward packets per second	$bpac\_psec = \frac{n_{bp}}{\sum_{i=1}^{n_{bp}} fiat[i]}$
31	Number of forward packets for the first 10 packets	$n_{\rm fp}^{(10)} = \sum_{i=1}^{10} (1 \text{ if } d[i] == 0 \text{ else } 0)$
32	Number of backward packets for the first 10 packets	$n_{\rm bp}^{(10)} = \sum_{i=1}^{10} (1 \text{ if } d[i] = 1 \text{ else } 0)$
33	Number of forward packets for the first 60 packets	$n_{\rm fp}^{(60)} = \sum_{i=1}^{10} (1 \text{ if } d[i] == 0 \text{ else } 0)$
34	Number of backward packets for the first 60 packets	$n_{\rm bp}^{(60)} = \sum_{i=1}^{10} (1 \text{ if } d[i] = 1 \text{ else } 0)$
35	Ratio of the number of backward packets to forward packets for the first 10 packets	$r_p^{(10)} = \frac{n_{\rm bp}^{(10)}}{n_{\rm fp}^{(10)}}$
36	Ratio of the number of backward packets to forward packets for the first 60 packets	$r_p^{(60)} = rac{n_{ m bp}^{(60)}}{n_{ m tp}^{(60)}}$

 Table 1. Description of candidate flow feature set.

**Definition 4** (Network flow). *Network flow is a set of packets with the same five-tuple (source IP, destination IP, source port, destination port, transport protocol) in the same network in a certain period of time.* 

# **Definition 5** (Bilateral flow). *Bilateral flow is a pair of flows having symmetrical source IP, source port, destination IP, and destination port in the same network in a certain period of time.*

In the previous studies, static features such as IP, port number, and TCP (Transmission Control Protocol) flags were used as part of the feature set [67], which reduces the robustness of the classification model. When the model is trained with the traffic of a certain network, it will not perform well on another network or in another period of the network. Considering portability and robustness for encrypted traffic classification, this study extracts only spatio-temporal statistical flow features, including the interval of packets, bytes of packets, count of packets, ratio of packets, and velocity of flow, instead of static features. These features are extracted from two directions of the entire flow (forward and backward) and four statistical dimensions (minimum, maximum, mean, and standard deviation). Table 1 shows that functions of min(), max(), mean() and std() are to obtain the minimum, maximum, average and standard deviation value of the array, respectively. *byte*[] is an array of the number of IP packet bytes, where *byte*[*i*] means the length of the *i*th IP packet bytes. *iat*[] is an array of interval arrival time between the two adjacent packets, where *iat*[*i*] means the time interval between the *i*th packet and the *i*+1th packet. *iatsd*[] is an array of the inter-arrival time of packets in the same direction. d[] is an array of the directions of packets. n,  $n_{fp}$ ,  $n_{bp}$  are the number of the all packets, forward packets and backward packets in a flow, respectively, where the maximum is 60. Because the extracted flow features are based on the first 60 packets of a flow at most, it decreases the calculation cost.

It is worth noting that when selecting the candidate feature set, this study adds the number and the proportion of packets from different directions of the first 10 or 60 packets of one flow as features, as indicated in features of 31–36. The first several packets of one flow always represent the important interaction or negotiation between the two hosts, which indicates the key characteristics of the traffic. Using these features can improve the accuracy of network traffic identification. The experiments in this study also prove that these features play a significant role in application type and content type identification for encrypted network flows.

#### 4.3. Fine-Grained Classification Method

Based on the developed framework and the WKNN algorithm, this section proposes a fine-grained classification method for encrypted traffic on the basis of the improved KNN algorithm (FCE-KNN), which is shown in Algorithm 3. FCE-KNN includes two parts: model training and real-time classification. Furthermore, several definitions related to the algorithm are given below.

**Definition 6** (FCE flow label). The label of the network flow in each layer of the fine-grained framework is called the FCE flow label. The ith layer label value of flow is  $L_i(flow)$ .

**Definition 7** (FCE feature index). *After training and selection, the feature indexes set used for real-time classification in the framework are denoted by FI.* 

Lines 1–5 are the offline model training part of FCE-KNN, which is based on the WKNN-Selfada algorithm. It sets the ratio of data division *raDiv* to 9, which means that the algorithm uses 90% of the training samples as the decision-sample set and the remaining 10% as the weight-update set in each training round. As the value of *nRound* does not exceed the ratio range from the size of the complete set to the size of the weights update training set, the model can avoid overfitting as much as possible. Overfitting is one of the main problems in machine learning. It means that the model completely learns the characteristics of the training data, but cannot generalize the laws in other data. Because of the right setting of *nRound* and *raDiv*, each training data would not be reused as a weight update sample and the characteristics of each weight update sample would just be learned once. Thus, it avoids the

over-learning situation that the model learns the characteristics of the same training data repeatedly, which may lead to difficulties of model generalization. Data normalization is a useful means to process data and is able to speed up the model training. In this paper, for a certain feature dimension, the new feature value of the *i*th data is given by  $x_i' = \frac{x_i - \mu(x)}{\sigma(x)}$ , where  $x_i$  is the old feature value;  $\mu(x)$  is the mean;  $\sigma(x)$  is the standard derivation; and  $x'_i$  is the new feature value.

Algor	thm 3: FCE-KNN									
	<b>Data</b> : $k_1, k_2, k_3, nRound_1, nRound_2, nRound_3, \delta_1, \delta_2, \delta_3, trainFlows$									
	Result: flow classes									
1	$raDiv \leftarrow 9$									
2	$trainFlows \leftarrow featureNormalization(trainFlows), \ enTrainFlows \leftarrow featureNormalization(enTrainFlows)$									
3	$w^{(1)}$ , $FI_1 \leftarrow WKNN\_Selfada(trainFlows, k_1, nRound_1, raDiv, \delta_1)$									
4	$w^{(2)}$ , $FI_2 \leftarrow WKNN\_Selfada(enTrainFlows, k_2, nRound_2, raDiv, \delta_2)$									
5	$w^{(3)}$ , $FI_3 \leftarrow WKNN\_Selfada(enTrainFlows, k_3, nRound_3, raDiv, \delta_3)$									
6	repeat(capture packets in real-time)									
7	$Flow^{(i)} \leftarrow pacMapToFlow(packet)$									
8	$len(Flow^{(i)}) + +$									
9	if $len(Flow^{(i)}) < 61$ then									
10	$Flow^{(i)} \leftarrow featureUpdate(Flow^{(i)}, len(packet), t(packet))$									
11	end									
12	if $len(Flow^{(i)}) == 60$ or $Flow^{(i)}$ is over then									
13	$Flow^{(i)} \leftarrow featureNormalization(Flow^{(i)})$									
14	$trainFlows_1, entrainFlows_2, entrainFlows_3 \leftarrow featureSelectProcessings(trainFlows, FI_1, FI_2, FI_3)$									
15	$Flow_1^{(i)}, Flow_2^{(i)}, Flow_3^{(i)} \leftarrow \text{featureSelectProcessings}(Flow^{(i)}, Fl_1, Fl_2, Fl_3)$									
16	$L_1(Flow^{(i)}) \leftarrow WKNN(trainFlows_1, Flow_1^{(i)}, k_1, w^{(1)})$									
17	if $L_1(Flow^{(i)})$ is encrypted flow then									
18	$L_2(Flow^{(i)}) \leftarrow WKNN(entrainFlows_2, Flow_2^{(i)}, k_2, w^{(2)})$									
19	$L_3(Flow^{(i)}) \leftarrow WKNN(entrainFlows_3, Flow_3^{(i)}, k_3, w^{(3)})$									
20	end									
21	end									
22	until traffic capture is stopped									

As shown in lines 3–5, the training part calculates the selected feature sets and feature weight sets of the sub-classifier models on the basis of the WKNN-Selfada algorithm. As for the parameters,  $k_1$ ,  $k_2$ ,  $k_3$ ;  $nRound_1$ ,  $nRound_2$ ,  $nRound_3$ ; and  $\delta_1$ ,  $\delta_2$ ,  $\delta_3$  are the values of k, nRound and  $\delta$  of the WKNN-Selfada algorithm implemented in each layer classification task, respectively. Furthermore, *trainFlows* represents the flow features of the training samples after data normalization, while *enTrainFlows* represents the encrypted flow data of *trainFlows*. The sets of  $FI_1$ ,  $FI_2$  and  $FI_3$  are the selected feature index, while  $w^{(1)}$ ,  $w^{(2)}$  and  $w^{(3)}$  are the feature weight sets after training.

When the training phase ends, all the sample data will be used as the decision samples in the online phase to calculate the point distance from the test point and determine its class.

Lines 6–22 are the real-time classification part of FCE-KNN, including real-time network flow mapping, flow feature real-time update, and flow label calculation based on WKNN. The model first extracts the five-tuple information, packet length, and timestamp of every online packet. The five-tuple

is used to map the online packet to a certain flow,  $Flow^{(i)}$ , and the length of  $Flow^{(i)}$  is increased by 1. The function featureUpdate() implements flow feature real-time update for  $Flow^{(i)}$  on the basis of the length of the packet, len(packet), and the timestamp, t(packet). As FCE-KNN extracts flow features from only the first 60 packets of each flow, the algorithm first judges whether the packet number of the flow reaches 60 in each instance. If the number does not exceed 60, the algorithm will use len(packet) and t(packet) of the processing packet to implement the flow feature update. Then, the algorithm will judge whether the packet number of the flow exceeds 60 or whether the flow is over. If so, the algorithm enters the label calculation part, as shown in lines 13–20. In this part, feature data normalization is the first operation to be implemented. Then, according to the selected feature index sets, trainFlows1, entrainFlows2 and entrainFlows3 are extracted as the training data for the different sub-models classification of the new flow, respectively. Similarly,  $Flow_1^{(i)}$ ,  $Flow_2^{(i)}$ ,  $Flow_3^{(i)}$ ,  $Flow_3^{(i$ are the extracted features of the target data, respectively. The first sub-classifier is used to identify the encryption status, i.e.,  $L_1(Flow^{(i)})$ . If it is an encrypted flow, the second and third sub-classifiers are used to identify the application type  $L_2(Flow^{(i)})$  and content type  $L_3(Flow^{(i)})$ , respectively. Finally, the algorithm outputs the fine-grained classification result. Note that if the classification result of the first sub-classifier is non-encrypted, the algorithm will directly output this classification result.

#### 5. Experiments and Evaluation

This section describes experiments based on the FCE-KNN method and compares the fine-grained classification performance of FCE-KNN with that of other, similar algorithms.

#### 5.1. Setup

The experimental platform was an MSI GT63 laptop with a six-core central processing unit (CPU, Intel Core i7-8750; 2.2 GHz) and 16 GB RAM. Experiments were performed on the public dataset ISCX VPN-nonVPN [16]. The public dataset used in this paper was captured in in-network routers, which was a representative dataset of real-world traffic generated by ISCX. The data in the dataset were raw traffic, without any preprocessing. The algorithms were implemented in Python.

The model needs to perform fine-grained analysis of application type and content type for encrypted traffic, and only the ISCX VPN-nonVPN dataset can meet the experimental requirements of this study on public datasets. Therefore, experiments were performed only on this dataset. The size of the ISCX VPN-nonVPN dataset is 25 GB, of which 22.8 G is plaintext traffic and the remaining 2.4 GB is encrypted traffic generated using VPN. The dataset contains 280,540 flows, of which 18,468 are ciphertext flows and 262,072 are plaintext flows, including 14 types of applications, covering a wide range of applications as well as multiple content types. Therefore, we used this dataset in our study.

To meet the experimental requirements, we added fine-grained classes for flows based on the dataset, so that the dataset can be used in the experiments. After class adjustment, each flow contains three classes (encryption status, application type, content type). The number of flows with different classes was summarized in Table 2. Note that in the classification tasks of the second and third layers, FCE-KNN only analyzes encrypted flows. The application label types include AIM (AmericanOnline Instant Messenger), ICQ (I Seek You, one of the message software), VoipBuster (one of the voice communication software), Spotify (one of the music service software), Hangouts (one of the message software), Youtube (one of the video communication software), SFTP (SSH File Transfer Protocol), Email, FTPS (File Transfer Protocol over SSL (Secure Sockets Layer)), Facebook (one of the message software), BitTorrent (one of the file transfer software), Netflix (one of the video communication software), Skype (one of the message software) and Vimeo (one of the message software). The content label types include chat, VoIP, file and streaming.

Label Type	Label Name and Number						
Encryption Status	Encry (184	Non-Encrypted (262072)					
	AIM (32)	Email (298)					
Application type	VoipBuster (1618)	FTPS (125) Facebook (2494)					
(Encrypted flows)	Spotify (137)	BitTorrent (477)					
(Encrypted nows)	Hangouts (10871)	Netflix (173)					
	YouTube (213)	Skype (1835)					
	SFTP (28)	Vimeo (136)					
Content type	Chat (4327)	File (1497)					
(Encrypted flows)	VoIP (11,985)	Streaming (659)					

Table 2. Flow labels and number of flows used in the experiments.

To test the performance of FCE-KNN, the dataset was divided into a train-validation set and test set. The test set was only used for the final performance test of the model. Meeting the need of parameters' validation, 10-fold cross-validation was used to improve the utilization of the train-validation set. To satisfy the requirement for the algorithms, the model training data of the train-validation set was further divided into a decision samples set and a weights update set in each fold of cross-validation. The details of the data division is shown in Figure 3.



Figure 3. Dataset division in experiments.

Although we have said that the analysis of the fine-grained classification of encrypted flows can be used for anomaly detection, we did not evaluate the performance. On the one hand, the main proposal of our research is to analyze encrypted network flows, but not to detect malicious flows, so we had not considered evaluating the anomaly detection effect of the presented method. On the other hand, anomaly detection based on the analysis of correlations between the flow attributes is a little subjective. There is not a suitable public dataset that can be used, and if we create a dataset that regards some flows with specific application and content to be malicious according to our judgement, and evaluate the presented method on this dataset, the results make no sense. This is because the malicious flows are designed by ourselves and we know the rules of anomaly in advance. Therefore, testing the performance of identifying the application and content type of encrypted flows is a more essential, useful and effective evaluation. As long as we identify the fine-grained attributes of the network flows, we can easily find out the malicious flows with strange combinations of application type and content type according to the network security rules.

#### 5.2. Metric

The following metrics were used to evaluate the classification effect: Accuracy, Precision, Recall, and F1-Score. The metrics are defined as:  $Accuracy = \frac{TP_i}{TP_i + FN_i}$ ,  $Precision = \frac{TP_i}{TP_i + FP_i}$ ,  $Recall = \frac{TP_i}{TP_i + FN_i}$  and  $F1 - score = \frac{2 \cdot Precision \cdot Recall}{Precision \cdot Recall}$ . Among them,  $TP_i$  is the number of *i*th type flows that are correctly classified,  $FP_i$  is the number of not *i*th type flows that are misclassified as *i*th type flows, and  $FN_i$  is the number of *i*th type flows that are misclassified as not *i*th type flows. In the first-layer classification, which was a two-class task, the types of flows are encrypted and non-encrypted. In the second and third layers of

classifications, i.e., multi-class tasks, the types of flows are different application (content), e.g., FTPS, Email etc. (Chat, File etc.).

#### 5.3. Experiments and Results

Some important parameters of FCE-KNN are k, nRound, and  $\delta$ . This section discusses the classification performance analysis of FCE-KNN and evaluates the optimal values of the parameters. According to the three classification layers in FCE-KNN, this section is divided into three subsections, including analyzing the performance of FCE-KNN in identifying the encryption status, application type, and content type of encrypted flows, respectively. The four state-of-the-art methods, including DTW-KNN (Dynamic Time warping based on KNN) [53], C4.5 (one of the decision trees) [16], ADA (ADAboost, one of the ensemble method) [32] and AISVM (Incremental Support Machine with Attenuation factor) [31] are compared with FCE-KNN. DTW-KNN [53] is a variant KNN, which is based on the optimization problem, dynamic time warping, with the minimum cumulative distance when the two templates are matching. ADA [32] is an ensemble machine learning algorithm, adaboost, which is an ensemble classifier with multiple simple C4.5. C4.5 [16] is one of the widely used decision tree algorithm. AISVM [31] is a modified version of the incremental support vector machine (ISVM), which introduces attenuation factor and improves accuracy. All the algorithms in comparison are network traffic-classification methods based on modified versions of machine-learning algorithms. Neural networks are also widely used in traffic classification, but they need large amount of training data, which make no sense in evaluation with a small training set.

#### 5.3.1. Identification of Encryption Status of Network Flows

This section verifies the performance of FCE-KNN in identifying the flow encryption status, i.e., whether a flow is an encrypted flow or not.

The optimal parameter values were first analyzed. As shown in Figure 4a,  $k_1$  is most likely to obtain an optimal value between 1 and 21. As shown in Figure 4b, when  $k_1$  is 5, the accuracy reaches the highest, which is 99.14%. The analysis of the optimal *nRound*<sub>1</sub> value is shown in Figure 5a. As *nRound*<sub>1</sub> increases, the verification accuracy of the model changes accordingly. In the 3rd and 8th rounds, the testing accuracy reaches its highest value without over-fitting. Considering that the three-round training requires a shorter training time, the best value for *nRound*<sub>1</sub> is 3. The analysis of the optimal  $\delta_1$  value is shown in Figure 5b. When the value is 0.6 or 0.8, the accuracy of the model is the highest, and as the value increases, the accuracy decreased significantly. Therefore, a value of 0.6–0.8 is relatively suitable. The experiment also outputs the feature weights in the feature selection part of the model training, as shown in Figure 6, where the red line indicates the feature selection threshold. It can be seen that the features of the packet interval and packet byte have a significant impact on the classification. The trained model retains only around 1/3 of all the features, further indicating that the reduction in traffic features can improve not only the efficiency of online feature extraction but also the classification accuracy.



**Figure 4.** Accuracy of fine-grained classification of encrypted network flows based on improved KNN (FCE-KNN) for identifying encryption status of network flows with different values of  $k_1$ : (a)  $k_1$  from 1 to 101, interval 10; (b)  $k_1$  from 1 to 20, interval 1.



**Figure 5.** Accuracy of FCE-KNN for identifying encryption status of network flows with different values of  $nRound_1$  or  $\delta_1$ :(**a**): Accuracy under different  $nRound_1$ ; (**b**) Accuracy under different  $\delta_1$ .



**Figure 6.** Comparison of feature weights in the feature selection part of the model training for identifying encryption status of network flows.

In this section, two sub-experiments were implemented. The first one was to train a model on the whole train-validation set and test the model on the test set. To verify the robustness of FCE-KNN for flow encryption status identification, the other one was performed to test whether the model can deal with the flows of an unknown application. In this experiment, assuming Hougout was an unknown application, we used all the data except Hougout flows and the corresponding encryption status labels to train a model, and tested the model on the all Hougout flows.

The parameter setting of FCE-KNN was as follows:  $k_1$  was set to 3,  $nRound_1$  was set to 3, and  $\delta_1$  was set to 0.6. As shown in Table 3, FCE-KNN exhibited the best performance. Although the performance of FCE-KNN was not significantly better than that of the other algorithms, the precision and recall of FCE-KNN were high in the case of unbalanced classes, which indicated that FCE-KNN was sensitive to encrypted flows. Note that it was not easy to maintain high precision and high recall at the same time. Thus, FCE-KNN has high adaptability to encrypted flows classification tasks. In addition, it performed best at identifying the encryption status of the unknown application flows, showing the best robustness of classification.

Method	Test on	Known-App	plication Flo	Test on Unknown-Application Flows				
memou	Acc (%)	Pre (%)	Rec (%)	F1	Acc (%)	Pre (%)	<b>Rec (%)</b>	F1
FCE-KNN	99.34	98.56	91.31	0.94	99.30	99.37	94.69	0.96
DTW-KNN [53]	98.99	96.71	87.68	0.91	98.20	94.46	90.12	0.92
C4.5 [16]	98.99	98.12	86.27	0.91	98.96	99.67	91.51	0.95
ADA [32]	99.11	97.87	88.41	0.92	99.06	99.99	92.08	0.95
AISVM [31]	95.01	59.17	78.01	0.67	88.10	45.94	0.31	0

**Table 3.** Performance comparison between FCE-KNN and other algorithms for identifying the encryption status of network flows.

Acc is Accuracy; Pre is Precision of identification of encrypted flows; Rec is Recall of identification of encrypted flows; F1 is F1-score of identification of encrypted flows.

#### 5.3.2. Identification of Application Type of Encrypted Network Flows

This section tests the performance of FCE-KNN in the application type classification of encrypted flows, i.e., what type of application is used. According to the dataset ISCX VPN-nonVPN, flows used in the experiment involve 14 types of applications: AIM, BitTorrent, email, Facebook, FTPS, Hangouts, ICQ, Netflix, SFTP, Skype, Spotify, Vimeo, VoipBuster, and YouTube.

First, paraments validation were implemented. As shown in Figure 7, the accuracy reaches its maximum value when  $k_2$  is equal to 11, and as  $k_2$  increased, the accuracy gradually decreases. When  $k_2$  is 8, the accuracy is the highest, which is 91.61%. The analysis of the optimal *nRound*<sub>2</sub> value is shown in Figure 8a. As *nRound*<sub>2</sub> increases, the validation accuracy of the model does not change significantly. It can be seen that the feature weights have been trained to converge in the first round; hence, the optimal value of *nRound*<sub>2</sub> is 1. The analysis of the best  $\delta_2$  value is shown in Figure 8b. The model achieves the highest accuracy of 91.68% at a value of -0.2. As the value increases, the accuracy decreases significantly. This is because there are only two or three features left in the selected feature set, which degraded the classification performance severely. Figure 9 shows all the feature weights in the feature selection part of the model training. The trained model retains only around half of the features, and the feature of packet byte plays an important role in the application-type identification of encrypted flows.



**Figure 7.** Accuracy of FCE-KNN for identifying application types of encrypted flows with different values of  $k_2$ : (**a**)  $k_2$  from 1 to 101, interval 10; (**b**)  $k_2$  from 1 to 20, interval 1.



**Figure 8.** Accuracy of FCE-KNN for identifying application types of encrypted flows with different values of *nRound*<sub>2</sub> or  $\delta_2$ :(**a**): Accuracy under different *nRound*<sub>2</sub>; (**b**) Accuracy under different  $\delta_2$ .



**Figure 9.** Comparison of feature weights in the feature selection part of model training for identifying application types of encrypted flows.

In this experiment, for the parameters of FCE-KNN,  $k_2$  was set to 8, *nRound*<sub>2</sub> was set to 1, and  $\delta_2$  was set to -0.2. As shown in Tables 4–7, FCE-KNN shows the best performance in the application

type identification of encrypted flows and the highest accuracy among the five algorithms, with an improvement of 2.4% over the second-ranked algorithm (DTW-KNN). AISVM performed well in Section 5.3.1, but it did not seem to perform well in multi-class tasks. In most of the classes, FCE-KNN has the highest Precision, Recall, and F1-score. Furthermore, we found that it was difficult to identify the flows of AIM, ICQ, and SFTP, mainly because the amount of flows of these classes was too small to train the model, whereas the number of training data of AIM, ICQ and SFTP were 24, 24 and 22, respectively. Nevertheless, FCE-KNN could correctly classify some of them such as AIM and ICQ, which performed better than other algorithms. Thus, FCE-KNN can rapidly adapt to the environment in the condition of imbalanced classes and small training set. Furthermore, note that FCE-KNN has the highest F1 scores in all the classes, which has fairly good stability, i.e., it balanced the precision and recall.

**Table 4.** Accuracy comparison between FCE-KNN and other algorithms for identifying application type of encrypted network flows.

Method	Accuracy (%)
FCE-KNN	92.45
DTW-KNN [53]	90.06
C4.5 [16]	85.71
ADA [32]	89.77
AISVM [31]	67.65

**Table 5.** Precision comparison between FCE-KNN and other algorithms for identifying application type of encrypted network flows.

Mathad	Precision (%)									
Method	AIM	BitTorrent	Email	Facebook	FTPS	Hangouts	ICQ 33.33 0 0 0 0 0 <b>YouTube</b> 83.33 70.73 0 80 0			
FCE-KNN	66.66	93.54	<b>3.54</b> 82.35 <b>89.13</b> 80.76 <b>96.40</b>		96.40	33.33				
DTW-KNN [53]	0	86.95	78.18	86.91	45.71	95.78	3 0			
C4.5 [16]	0	77.77	0	96.06	0	95.04	0			
ADA [32]	0	85 <b>87.5</b> 82.79 <b>84.61</b> 95.39		95.39	0					
AISVM [31]	0	0	0 0 0		72.02	0				
	Precision (%)									
Method	Netflix	SFTP	Skype	Spotify	Vimeo	VoipBuster	YouTube			
FCE-KNN	52.5	0	83.84	45.45	50	98.15	83.33			
DTW-KNN [53]	29.41	0	76.22	35.29	60	98.70	70.73			
C4.5 [16]	62.5	0	47.39	41.66	0	95.93	0			
ADA [32]	59.25	0	72.41	57.14	32.60	97.22	80			
AISVM [31]	0	0	28.69	0	0	84.59	0			

Mathad	Recall (%)									
Method	AIM	BitTorrent	Email	Facebook	FTPS	Hangouts	ICQ			
FCE-KNN	33.33	<b>91.57 93.33</b> 87.17 <b>84 9</b>		97.33	16.66					
DTW-KNN [53]	0	84.21	71.66	85.17	64	97.19	0			
C4.5 [16]	0	66.31	0	73.34	0	96.27	0			
ADA [32]	0	71.57	70	87.77	44	97.24	0			
AISVM [31]	0	0	0	0	0 0 96.18		0			
	Recall (%)									
Method	Netflix	SFTP	Skype	Spotify	Vimeo	VoipBuster	YouTube			
FCE-KNN	60	0	82.01	37.03	59.25	98.45	69.76			
DTW-KNN [53]	42.85	0	76.02	44.44	33.33	94.44	67.44			
C4.5 [16]	14.28	0	89.10	18.51	0	94.75	0			
ADA [32]	45.71	0	74.38	29.62	55.55	97.22	37.20			
AISVM [31]	0	0	34.87	0	0	86.41	0			

**Table 6.** Recall comparison between FCE-KNN and other algorithms for identifying application type of encrypted network flows.

**Table 7.** F1-score comparison of FCE-KNN for identifying application type of encrypted network flows with other algorithms.

Madaal	F1-Score									
Method	Aim	Bittorrent	Email	Facebook	Ftps	Hangouts	Icq 0.22 0 0 0 0 0 0 <b>Youtube</b> 0.75 0.69 0 0.50			
FCE-KNN	0.44	0.92	0.87	0.88	0.82	0.96	0.22			
DTW-KNN [53]	0	0.85	0.74	0.86	0.53	0.96	0			
C4.5 [16]	0	0.71	0	0.83	0	0.95	0			
ADA [32]	0	0.77	0.77	0.85	0.57	0.96	0			
AISVM [31]	0	0	0 0 0 0.83		0.83	0				
Mathad	F1-Score									
Method	Netflix	Sftp	Skype	Spotify	Vimeo	Voipbuster	Youtube			
FCE-KNN	0.56	0	0.82	0.40	0.54	0.98	0.75			
DTW-KNN [53]	0.34	0	0.76	0.39	0.42	0.96	0.69			
C4.5 [16]	0.23	0	0.61	0.25	0	0.95	0			
ADA [32]	0.51	0	0.73	0.39	0.41	0.97	0.50			
AISVM [31]	0	0	0.31	0	0	0.85	0			

#### 5.3.3. Identification of Content Type of Encrypted Network Flows

This section tests the performance of FCE-KNN in identifying the content type of encrypted flows, i.e., what type of content is carried by the encrypted flows. According to the public dataset ISCX VPN-nonVPN and subsequent manual marking, the flows used in the experiment include four types of content: chat, file, VoIP, and streaming.

First, paraments validation were implemented. As shown in Figure 10, the accuracy reaches its maximum value when  $k_3$  is equal to 11, and as  $k_3$  increases, the accuracy gradually decreases. When  $k_3$  is 1, the accuracy is the highest, which is 96.67%. The analysis of the optimal *nRound*<sub>3</sub> value is shown in Figure 11a. As *nRound*<sub>3</sub> increases, the accuracy of the model does not change significantly. It can be seen that feature weights have been trained to converge in the first round; hence, the optimal value of *nRound*<sub>3</sub> is 1. The analysis of the best  $\delta_3$  value is shown in Figure 11b. The model has the highest testing accuracy when  $\delta_3$  is 0, where the feature selection threshold is the mean of all the feature weights. Figure 12 shows all the feature weights in the feature selection part of the model training. The trained model retains only around half of the features, and the features of packet interval and packet byte have significant influence on the model classification.



**Figure 10.** Accuracy of FCE-KNN for identifying content type of encrypted flows with different values of  $k_3$ : (**a**)  $k_3$  from 1 to 101, interval 10; (**b**)  $k_3$  from 1 to 20, interval 1.







**Figure 12.** Comparison of feature weights in the feature selection part of model training for identifying content type of encrypted flows.

The performances of FCE-KNN and the other algorithms are summarized in Tables 8 and 9. The parameter setting of FCE-KNN was as follows:  $k_3$  was set to 1, *nRound*<sub>3</sub> was set to 1, and  $\delta_3$  was set to 0. It can be seen that the FCE-KNN also has the highest accuracy in the content type identification of encrypted flows, with an improvement of 2.7% over the second-ranked algorithm (ADA), and it has the highest F1-score in all the classes, indicating that FCE-KNN has high adaptability in identifying the content type of encrypted flows. Note that for the prediction of streaming flows, the performance of other algorithms are obviously lower than FCE-KNN by approximately 10%. The reason may be that the streaming flows does not have strong regularity, suggesting that FCE-KNN has excellent learning ability even if the data do not have strong regularity.

**Table 8.** Accuracy comparison between FCE-KNN and other algorithms for identifying content type of encrypted network flows.

Method	Accuracy (%)
FCE-KNN	97.02
DTW-KNN [53]	94.50
C4.5 [16]	92.69
ADA [32]	95.37
AISVM [31]	66.00

**Table 9.** Performance comparison between FCE-KNN and other algorithms for identifying content type of encrypted network flows.

Mathad	Chat			File		Streaming			VoIP			
Method	P(%)	R(%)	F1	P(%)	R(%)	F1	P(%)	R(%)	F1	P(%)	R(%)	F1
FCE-KNN	95.78	96.99	0.96	88.85	90.63	0.89	91.72	92.42	0.92	98.82	98.08	0.98
DTW-KNN [53]	93.68	94.22	0.93	82.43	81.60	0.82	80.62	78.78	0.79	97.03	97.07	0.97
C4.5 [16]	97.64	91.10	0.94	64.10	83.61	0.72	82.14	52.27	0.63	96.01	96.62	0.96
ADA [32]	97.04	94.80	0.95	80.93	80.93	0.80	83.33	75.75	0.79	97.15	98.45	0.97
AISVM [31]	0	0	0	50.30	27.42	0.35	0	0	0	66.74	98.28	0.79

P is Precision; R is Recall; F1 is F1-score.

#### 5.4. Analysis of Time Complexity and Consumption

It is important to analyze the complexity and time consumption of an algorithm. Low time-consumption is a significant requirement for real-time classification. FCE-KNN consists of WKNN and WKNN-Selfada algorithms. We first analyzed the time complexity of the algorithms. For WKNN, to predict the label of the test point, the algorithm will calculate the point distance between the test point and each decision training sample, so the time complexity is  $O(m) \approx O(n)$ , where m is the size of the decision sample set, and n is the size of the whole training set. For WKNN-Selfada, to select the features and obtain the corresponding feature weights, the algorithm will first calculate the point distance between the each weight's update sample and each decision training sample, so the time complexity is  $O(m \cdot (n - m)) \approx O(n^2)$ , where n - m is the size of the weights update set. Accordingly, the training time complexity of FCE-KNN is  $O(n^2)$ , and the test time complexity is O(n). In addition, the training time complexities of DTW-KNN, C4.5, ADA and AISVM are  $O(n^2)$ ,  $O(\log(n) \cdot n)$ ,  $O(\log(n) \cdot n \cdot k)$ ,  $O(n^2)$ , respectively, and the test time complexities are O(n), O(d),  $O(d \cdot k)$ , O(s), respectively, where k is the number of trees, d is the depth of the trees, and s is the number of the support vectors.

Then the actual time consumption is analyzed, as shown in Table 10. The training numbers of the flows in the experiments of the three layers are 224,432; 14,774; and 14,774, respectively. The test numbers of the flows in the experiments of the three layers are 26,108; 3694; and 3694, respectively. The sizes of the selected feature sets of the three layers are 12, 19, 18. It could be seen that the time consumptions of FCE-KNN are a little high, just lower than ADA and AISVM. It is

possible that FCE-KNN processes the feature selection and feature weights calculation, which is more time-consuming than traditional KNN. It is also because the algorithm learns the characteristics from each sample completely, consuming more time to train on a single sample. Thus, the total time consumption is a little higher than similar algorithms.

Method	Training			Testing		
	1st Layer	2nd Layer	3rd Layer	1st Layer	2nd Layer	3rd Layer
FCE-KNN	116.33	0.31	0.25	16.98	0.22	0.22
DTW-KNN [53]	77.15	0.20	0.17	15.71	0.20	0.19
C4.5 [16]	0.98	0.04	0.04	0.01	0.01	0.01
ADA [32]	206.93	10.75	9.45	0.73	0.15	0.07
AISVM [31]	1595.10	9.75	10.00	67.51	1.61	1.41

**Table 10.** Time consumption comparison between FCE-KNN and other algorithms in the framework of fine-grained classification (s).

#### 5.5. Discussion

It can be seen from the experimental results in Section 5.3 that FCE-KNN is not sensitive to the *k* value. FCE-KNN overcomes the drawbacks of the traditional KNN algorithm, i.e., it is more robust and shows better performance. Furthermore, FCE-KNN is not sensitive to *nRound*. This may be because the weight update samples are sufficient and the feature weights converge after the first round of training. For the threshold  $\delta$  used for feature selection, the performance is stable before reaching the optimum value, and the accuracy of the model will decrease rapidly after a certain peak value. This may be because the size of the selected feature set will be extremely small, and there may be only two or three features when the value is large. From the curve of  $\delta$ , we can conclude that for traffic classification the feature set should not be as large as possible; redundant features may degrade the classification performance owing to the weak correlation with the classification target.

The performance of FCE-KNN is not always optimal in all the experiments. Although, the performance is not the best in a few cases of identification of a single class, the F1 scores are the highest in all the classes, indicating that FCE-KNN is a highly balanced algorithm with both high precision and high recall. Moreover, FCE-KNN has high practicability as it can identify the encryption status of unknown applications. In the experiments described in Sections 5.3.2 and 5.3.3, FCE-KNN shows better performance than the other algorithms in the case of class imbalance, such as small amounts of ICQ, AIM, and Netflix flows in the task for identifying the application type of encrypted flows. For example, under the situation that the number of AIM flows was 32, the precision of FCE-KNN was 66%, while other algorithms were 0. To some extent, FCE-KNN alleviates the problem of class imbalance and small training set and shows strong adaptability to actual traffic environments. From the experimental results, it is seen that FCE-KNN performs better than those algorithms because different features have different influences upon different classification tasks and the presented algorithm has expressed the different influences by introducing feature weights self-adapting adjustment. It is proved that the thought of feature selection and feature weights self-adaption are effective and feasible.

As for the fact that FCE-KNN can train an accurate model with a small training set, we can evaluate this from two perspectives. On the one hand, the WKNN-Selfada algorithm selectes features by comparing the trained feature weights, and the feature weights are updated by a single training sample each time, not a batch. So it could fully learn the characteristics of each sample and train an accurate model just by using a small training set. This means that the algorithm adjusts the feature weights on each sample. In other words, every sample is beneficial and useful for the model training, which is different from some algorithms in which the model is trained on some special samples. On the other hand, in the experiments of identification of application type and content type of network flows, the number of some flows is very small, such as AIM, ICQ, FTPS, of which the numbers are 32, 31, 125,

respectively. For those flows with a small amount, FCE-KNN shows obvious superiority comparing with other algorithms, which proves that FCE-KNN is able to train a model in the case of a small training set.

However, time consumption of FCE-KNN is a little higher than the other algorithms, which means more good hardware is required to achieve real-time classification. Owing to processes of feature selection and feature weights self-adaption, the time consumption of model training is high, but this did not affect the real-time processing. The feature extraction and distance calculation can be incrementally updated, where the raw packets do not need to be stored and the packet data can just be processed once. To sum up, FCE-KNN improves the traditional KNN algorithm and performs better than similar algorithms in accuracy, while it is a little time-consuming.

#### 6. Conclusions

This study improves the traditional KNN algorithm and adopts it for the fine-grained classification of encrypted network flows. The experimental results verify the feasibility of the improved algorithm, which not only outperforms the traditional KNN algorithm but also shows stronger stability and higher performance than other similar methods. Furthermore, feature selection based on feature weights and point distance calculation is shown to be effective. The proposed WKNN-Selfada algorithm can be applied to actual traffic environments after it is trained to learn the laws of network flows.

However, this study still has the following limitations. FCE-KNN is dependent on training samples and it is more time-consuming than other algorithms such as decision trees. Therefore, in the future, we will try to exploit other machine-learning algorithms or establish a hybrid model that combines the advantages of different methods, and apply them to the fine-grained classification framework.

**Author Contributions:** C.M. designed the improved algorithms and the fine-grained framework, evaluated the method proposed through experiments and writing the manuscript. X.D. and L.C. reviewed and modified the article. All authors have read and agreed to the published version of the manuscript.

**Funding:** This work was supported in part by the National Key Research and Development Program of China under Grant 2018YFB0803603 and Grant 2016YFB0501901, and in part by the National Natural Science Foundation of China under Grant 61502531, Grant 61702550, and Grant 61802436.

Acknowledgments: We thank ISCX for free public datasets for our research, where we implement the experiments and evaluation on their public dataset, ISCX VPN-nonVPN.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- Dhote, Y.; Agrawal, S.; Deen, A.J. A Survey on Feature Selection Techniques for Internet Traffic Classification. In Proceedings of the International Conference on Computational Intelligence & Communication Networks (CICN), Madhya Pradesh, India, 12–14 December 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 1375–1380.
- Rathore, M.M.; Ahmad, A.; Paul, A.; Rho, S. Exploiting encrypted and tunneled multimedia calls in high-speed big data environment. *Multimed. Tools Appl.* 2018, 77, 4959–4984. [CrossRef]
- Velan, P.; Čermák, M.; Čeleda, P.; Drašar, M. A survey of methods for encrypted traffic classification and analysis. *Int. J. Netw. Manag.* 2015, 25, 355–374. [CrossRef]
- Hirvonen, M.; Sailio, M. Two-Phased Method for Identifying SSH Encrypted Application Flows. In Proceedings of the 7th International Wireless Communications and Mobile Computing Conference (IWCMC), Istanbul, Turkey, 4–8 July 2011; IEEE: Piscataway, NJ, USA, 2011; pp. 1033–1038.
- 5. Wang, Z. The applications of deep learning on traffic identification. *BlackHat* 2015, 24, 1–10.
- 6. Tan, X.; Xie, Y.; Ma, H.; Yu, S.; Hu, J. Recognizing the content types of network traffic based on a hybrid DNN-HMM model. *J. Netw. Comput. Appl.* **2019**, *142*, 51–62. [CrossRef]
- Cao, Z.; Xiong, G.; Zhao, Y.; Li, Z.; Guo, L. A Survey on Encrypted Traffic Classification. In Proceedings of the 2014 International Conference on Applications and Techniques in Information Security (ATIS), Melbourne, VIC, Australia, 26–28 November 2014; Springer: Berlin/Heidelberg, Germany, 2014; pp. 73–81.

- Alshammari, R.; Zincir-Heywood, A.N. Machine Learning Based Encrypted Traffic Classification: Identifying ssh and Skype. In Proceedings of the 2009 IEEE Symposium on Computational Intelligence for Security and Defense Applications (CISDA), Ottawa, ON, Canada, 8–10 July 2009; IEEE: Piscataway, NJ, USA, 2009; pp. 1–8.
- 9. Usama, M.; Qadir, J.; Raza, A.; Arif, H.; Yau, K.A.; Elkhatib, Y.; Hussain, A.; Al-Fuqaha, A. Unsupervised machine learning for networking: Techniques, applications and research challenges. *IEEE Access* 2019, 7, 65579–65615. [CrossRef]
- 10. Limthong, K.; Fukuda, K.; Ji, Y.; Yamada, S. Unsupervised learning model for real-time anomaly detection in computer networks. *IEICE Trans. Inf. Syst.* **2014**, *97*, 2084–2094. [CrossRef]
- 11. Suthaharan, S. Big data classification: Problems and challenges in network intrusion prediction with machine learning. *ACM Signetrics Perform. Eval. Rev.* **2014**, *41*, 70–73. [CrossRef]
- 12. Krawczyk, B.; Minku, L.L.; Gama, J.; Stefanowski, J.; Wozniak, M. Ensemble learning for data stream analysis: A survey. *Inf. Fusion* **2017**, *37*, 132–156. [CrossRef]
- 13. Thanh Noi, P.; Kappas, M. Comparison of random forest, k-nearest neighbor, and support vector machine classifiers for land cover classification using Sentinel-2 imagery. *Sensors* **2018**, *18*, 18. [CrossRef]
- Oehmcke, S.; Zielinski, O.; Kramer, O. KNN Ensembles with Penalized DTW for Multivariate Time Series Imputation. In Proceedings of the 2016 International Joint Conference on Neural Networks (IJCNN), Vancouver, BC, Canada, 24–29 July 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 2774–2781.
- 15. Liu, Z.; Wang, R.; Japkowicz, N.; Cai, Y.; Tang, D.; Cai, X. Mobile app traffic flow feature extraction and selection for improving classification robustness. *J. Netw. Comput. Appl.* **2019**, *125*, 190–208. [CrossRef]
- Draper-Gil, G.; Lashkari, A.H.; Mamun, M.S.I.; Ghorbani, A.A. Characterization of Encrypted and Vpn Traffic Using Time-Related. In Proceedings of the 2nd International Conference on Information Systems Security and Privacy (ICISSP), Rome, Italy, 19–21 February 2016; pp. 407–414.
- Yin, C.; Wang, H.; Wang, J. Network Data Stream Classification by Deep Packet Inspection and Machine Learning. In *Advanced Multimedia and Ubiquitous Engineering, Proceedings of the FutureTech 2018, Salerno, Italy,* 23–25 April 2018; Springer: Berlin/Heidelberg, Germany, 2018; pp. 245–251.
- Sherry, J.; Lan, C.; Popa, R.A.; Ratnasamy, S. Blindbox: Deep packet inspection over encrypted traffic. ACM SIGCOMM Comput. Commun. Rev. 2015, 45, 213–226. [CrossRef]
- Meiners, C.; Norige, E.; Liu, A.X.; Torng, E. Flowsifter: A counting Automata Approach to Layer 7 Field Extraction for Deep Flow Inspection. In Proceedings of the IEEE INFOCOM, Orlando, FL, USA, 25–30 March 2012; IEEE: Piscataway, NJ, USA, 2012; pp. 1746–1754.
- 20. Zeng, X.; Chen, X.; Shao, G.; He, T.; Han, Z.; Wen, Y.; Wang, Q. Flow Context and Host Behavior Based Shadowsocks's Traffic Identification. *IEEE Access* **2019**, *7*, 41017–41032. [CrossRef]
- 21. Zhu, H.; Zhu, L. Online and automatic identification of encryption network behaviors in big data environment. *Pract. Exp.* **2019**, *31*, e4849.
- Zygmunt, M.; Konieczny, M.; Zielinski, S. Accuracy of Statistical Machine Learning Methods in Identifying Client Behavior Patterns at Network Edge. In Proceedings of the 42nd International Conference on Telecommunications and Signal Processing (TSP), Budapest, Hungary, 3–5 July 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 575–579.
- 23. An, H.M.; Lee, S.K.; Ham, J.H.; Kin, M. Traffic Identification Based on Applications using Statistical Signature Free from Abnormal TCP Behavior. *J. Inf. Sci. Eng.* **2015**, *31*, 1669–1692.
- 24. Wang, H.; Qian, C.; Yu, Y.; Yang, H.; Lam, S.S. Practical network-wide packet behavior identification by AP classifier. *IEEE/ACM Trans. Netw.* **2017**, *25*, 2886–2899. [CrossRef]
- Zhu, A. A P2P Network Traffic Classification Method Based on C4. 5 Decision Tree Algorithm. In Proceedings of the 9th International Symposium on Linear Drives for Industry Applications (LDIA), Hangzhou, China, 7–10 January 2013; Springer: Berlin/Heidelberg, Germany, 2014; pp. 373–379.
- Linping, S.; Hongtao, M.; Yunlang, M.; Rui, C. The Research of Classified Method of the Network Traffic in Security Access Platform Based on Decision Tree. In Proceedings of the 7th IEEE International Conference on Software Engineering & Service Science (ISESS), Beijing, China, 26–28 August 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 475–480.
- 27. Fries, T.P. Classification of Network Traffic Using Fuzzy Clustering for Network Security. In Proceedings of the Industrial Conference on Data Mining (ICDM), New York, NY, USA, 12–13 July 2017; Volume 1, pp. 278–285.

- 28. Kim, J.; Sim, A. A New Approach to Multivariate Network Traffic Analysis. J. Comput. Sci. Technol. 2019, 34, 388–402. [CrossRef]
- 29. Cha, S.; Kim, H. Detecting Encrypted Traffic: A Machine Learning Approach. In Proceedings of the International Workshop on Information Security Application (WISA), Jeju Island, Korea, 25–27 August 2016; pp. 54–65.
- 30. Vargas-Muñoz, M.J.; Martínez-Peláez, R.; Velarde-Alvarado, P.; Moreno-García, E.; Torres-Roman, D.L.; Ceballos-Mejía, J.J. Classification of Network Anomalies in Flow Level Network Traffic Using Bayesian Networks. In Proceedings of the 2018 International Conference on Electronics, Communications and Computers, Cholula Puebla, Mexico, 21–23 February 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 238–243.
- Sun, G.; Chen, T.; Su, Y.; Li, C. Internet traffic classification based on incremental support vector machines. *Mob. Netw. Appl.* 2018, 23, 789–796. [CrossRef]
- 32. Gómez, S.E.; Martínez, B.C.; Sánchez-Esguevillas, A.J.; Callejo, L.H. Ensemble network traffic classification: Algorithm comparison and novel ensemble scheme proposal. *Comput. Netw.* **2017**, 127, 68–80. [CrossRef]
- De Souza, E.N.; Matwin, S.; Fernandes, S. Network Traffic Classification Using AdaBoost Dynamic. In Proceedings of the IEEE International Conference on Communications Workshops (ICC), Budapest, Hungary, 9–13 June 2013; IEEE: Piscataway, NJ, USA, 2013; pp. 1319–1324.
- 34. Yang, Y.; Kang, C.; Gou, G.; Li, Z.; Xiong, G. TLS/SSL Encrypted Traffic Classification with Autoencoder and Convolutional Neural Network. In Proceedings of the IEEE 20th International Conference on High Performance Computing and Communications; IEEE 16th International Conference on Smart City; IEEE 4th International Conference on Data Science and Systems (HPCC/SmartCity/DSS), Mercure Exeter, UK, 28–30 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 362–369.
- 35. Zou, Z.; Ge, J.; Zheng, H.; Han, C.; Yao, Z. Encrypted Traffic Classification with a Convolutional Long Short-Term Memory Neural Network. In Proceedings of the IEEE 20th International Conference on High Performance Computing and Communications; IEEE 16th International Conference on Smart City; IEEE 4th International Conference on Data Science and Systems (HPCC/SmartCity/DSS), Mercure Exeter, UK, 28–30 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 329–334.
- 36. Aceto, G.; Ciuonzo, D.; Montieri, A.; Pescape, A. MIMETIC: Mobile encrypted traffic classification using multimodal deep learning. *Comput. Netw.* **2019**, *165*, 106944. [CrossRef]
- Aceto, G.; Ciuonzo, D.; Montieri, A.; Pescape, A. Mobile encrypted traffic classification using deep learning: Experimental evaluation, lessons learned, and challenges. *IEEE Trans. Netw. Serv. Manag.* 2019, 16, 445–458. [CrossRef]
- 38. Aceto, G.; Ciuonzo, D.; Montieri, A.; Pescape, A. Multi-classification approaches for classifying mobile app traffic. *J. Netw. Comput. Appl.* **2018**, *103*, 131–145. [CrossRef]
- Aceto, G.; Ciuonzo, D.; Montieri, A.; Pescape, A. Mobile encrypted traffic classification using deep learning. In Proceedings of the Network Traffic Measurement and Analysis Conference (TMA), Austria, Vienna, 26–29 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–8.
- 40. Lotfollahi, M.; Siavoshani, M.J.; Zade, R.S.H.; Saberian, M. Deep packet: A novel approach for encrypted traffic classification using deep learning. *Soft Comput.* **2020**, *24*, 1999–2012. [CrossRef]
- 41. Zeng, Y.; Gu, H.; Wei, W.; Guo, Y. Deep-Full-Range: A Deep Learning Based Network Encrypted Traffic Classification and Intrusion Detection Framework. *IEEE Access* **2019**, *7*, 45182–45190. [CrossRef]
- Song, M.; Ran, J.; Li, S. Encrypted Traffic Classification Based on Text Convolution Neural Networks. In Proceedings of the IEEE 7th International Conference on Computer Science and Network Technology (ICCSNT), Dalian, China, 19–20 October 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 432–436.
- 43. Rezaei, S.; Liu, X. Multitask Learning for Network Traffic Classification. Available online: https://arxiv.org/ abs/1906.05248 (accessed on 10 February 2020).
- 44. Sun, G.; Liang, L.; Chen, T.; Xiao, F.; Lang, F. Network traffic classification based on transfer learning. *Comput. Electr. Eng.* **2018**, *69*, 920–927. [CrossRef]
- Sun, G.; Li, S.; Chen, T.; Li, X.; Zhu, S. Active Learning Method for Chinese Spam Filtering. *Int. J. Perform. Eng.* 2017, 13, 511–518. [CrossRef]
- 46. Zhu, H.; Zhu, L.; Shen, M.; Khan, S. Online and automatic identification and mining of encryption network behavior in big data environment. *J. Intell. Fuzzy Syst.* **2018**, *34*, 1111–1119.

- Wang, W.; Zhu, M.; Wang, J.; Zeng, X.; Yang, Z. End-to-End Encrypted Traffic Classification with One-Dimensional Convolution Neural Networks. In Proceedings of the IEEE International Conference on Intelligence and Security Informatics (ISI), Beijing, China, 22–24 July 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 43–48.
- 48. Wang, W.; Zhu, M.; Zeng, X.; Ye, X.; Sheng, Y. Malware Traffic Classification Using Convolutional Neural Network for Representation Learning. In Proceedings of the International Conference on Information Networking (ICOIN), Da Nang, Vietnam, 11–13 January 2017; pp. 712–717.
- Xia, J.; Shen, J.; Wu, Y. A Four-Stage Hybrid Feature Subset Selection Approach for Network Traffic Classification Based on Full Coverage. In Proceedings of the International Conference on Security, Privacy and Anonymity in Computation, Communication and Storage (SpaCCS), Guangzhou, China, 12–15 December 2017; Springer: Berlin/Heidelberg, Germany; pp. 178–191.
- Dorfinger, P.; Panholzer, G.; John, W. Entropy Estimation for Real-Time Encrypted Traffic Identification (Short Paper). In Proceedings of the International Workshop on Traffic Monitoring and Analysis (TMA), Vienna, Australia, 27–30 April 2011; Springer: Berlin/Heidelberg, Germany; pp. 164–171.
- Shen, M.; Zhang, J.; Zhu, L.; Xu, K.; Du, X.; Liu, Y. Encrypted Traffic Classification of Decentralized Applications on Ethereum Using Feature Fusion. In Proceedings of the IEEE/ACM International Symposium on Quality of Service (IWQoS), Phoenix, AZ, USA, 24–28 June 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 180–190.
- 52. Wu, D.; Chen, X.; Chen, C.; Zhang, J.; Xiang, Y.; Zhou, W. On Addressing the Imbalance Problem: A Correlated KNN Approach for Network Traffic Classification. In Proceedings of the International Conference on Network and System Security (NSS), New York, NY, USA, 3–5 November 2015; pp. 138–151.
- 53. Zhu, H.; Zhu, L. Encrypted network behaviors identification based on dynamic time warping and k-nearest neighbor. *Clust. Comput.* **2017**, *20*, 1–10. [CrossRef]
- Carela-Español, V.; Barlet-Ros, P.; Solé-Simó, M.; Dainotti, A.; Donato, W.D.; Pescape, A. K-Dimensional Trees for Continuous Traffic Classification. In Proceedings of the Second International Workshop on Traffic Monitoring and Analysis, Zurich, Switzerland, 7 April 2010; Springer: Berlin/Heidelberg, Germany, 2010; pp. 1–14.
- 55. Bar-Yanai, R.; Langberg, M.; Peleg, D.; Roditty, L. Realtime Classification for Encrypted Traffic. In Proceedings of the 9th International Symposium on Experimental Algorithms, Ischia Island, Naples, Italy, 20–22 May 2010; Springer: Berlin/Heidelberg, Germany, 2010; pp. 373–385.
- 56. McGaughey, D.; Semeniuk, T.; Smith, R.; Knight, S. A systematic approach of feature selection for encrypted network traffic classification. In Proceedings of the IEEE International Systems Conference (SysCon), Vancouver, BC, Canada, 23–26 April 2018; pp. 1–8.
- 57. Dong, Y.; Cao, R.; Zhang, M. A Multi-Objective Evolutionary Algorithm for Multimedia Traffic Classification. In Proceedings of the 2019 IEEE 21st International Conference on High Performance Computing and Communications; IEEE 17th International Conference on Smart City; IEEE 5th International Conference on Data Science and Systems (HPCC/SmartCity/DSS), Zhangjiajie, China, 10–12 August 2019; IEEE: Piscataway, NJ, USA, 2018; pp. 2804–2810.
- Saber, A.; Belkacem, F.; Moncef, A. Encrypted Network Traffic Identification: LDA-KNN Approach. In Proceedings of the 9 ème édition du colloque Tendances dans les Applications Mathématiques en Tunisie Algérie et Maroc, Tlemcen, Algeria, 23–27 February 2019; pp. 1–3.
- 59. Manju, N.; Harish, B.S.; Prajwal, V. Ensemble Feature Selection and Classification of Internet Traffic using XGBoost Classifier. *Int. J. Comput. Netw. Inf. Secur.* **2019**, *11*, 37.
- Jamil, H.A. Feature Selection and Machine Learning Classification for Live P2P Traffic. In Proceedings of the International Conference on Industrial Engineering and Operations Management (IEOM), Bangkok, Thailand, 5–7 March 2019; pp. 1–9.
- 61. Bugata, P.; Drotár, P. Weighted nearest neighbors feature selection. *Knowl. Based Syst.* **2019**, *163*, 749–761. [CrossRef]
- 62. Sun, B.; Cheng, W.; Goswami, P.; Bai, G. Short-term traffic forecasting using self-adjusting k-nearest neighbors. *IET Intell. Transp. Syst.* 2017, 12, 41–48. [CrossRef]
- 63. Saleh, A.I.; Talaat, F.M.; Labib, L.M. A hybrid intrusion detection system (HIDS) based on prioritized k-nearest neighbors and optimized SVM classifiers. *Artif. Intell. Rev.* **2019**, *51*, 403–443. [CrossRef]

- 64. Su, M. Using clustering to improve the KNN-based classifiers for online anomaly network traffic identification. *J. Netw. Comput. Appl.* **2011**, *34*, 722–730. [CrossRef]
- 65. Ma, Y.; Xie, Q.; Liu, Y.; Xiong, S. A weighted KNN-based automatic image annotation method. *Neural Comput. Appl.* **2019**, 1–12. [CrossRef]
- 66. Dong, Y.; Zhao, J.; Jin, J. Novel feature selection and classification of Internet video traffic based on a hierarchical scheme. *Comput. Netw.* **2017**, *119*, 102–111. [CrossRef]
- 67. Alshammari, R.; Zincir-Heywood, A.N. Can encrypted traffic be identified without port numbers, IP addresses and payload inspection? *Comput. Netw.* **2011**, *55*, 1326–1350. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).